Author's response to reviews

Title: Accuracy of portrayal by standardized patients: Results from a 10 station OSCE conducted for high stakes examinations.

1. Lubna A. Baig Corresponding Author Email: lbaig@ucalgary.ca and baigl@ksau-hs.edu.sa 4755 Hamptons Way NW Calgary, AB # T3A 6K1, Canada 2. Tanya N Beran Email: tnaberan@ucalgary.ca 3. Andrea Vellavand Email: avalleva@ucalgary.ca 4. Zarrukh Ali Baig Email: zabaig@ucalgary.ca 5. Mauricio Monroy-Cuadros Email: mmonroy@ucalgary.ca

Authors:

- Lubna A Baig (baigl@ksau-hs.edu.sa)
- Tanya N Beran (tnaberan@ucalgary.ca)
- Andrea Vellavand (avalleva@ucalgary.ca)
- Zarrukh A Baig (zabaig@ucalgary.ca)
- Mauricio P Monroy-Cuadros (mmonroy@ucalgary.ca)

Version: 2

Date: 10 December 2013

Author's response to reviews:

Title page now includes email addresses of all the authors and full address of the corresponding authors. The section on competing interests and authors contribution has been added to the manuscript.